Author's response to reviews

Title: Towards real-time metabolic profiling of biopsy specimen during a surgical operation by 1H HRMAS-NMR: a case report

Authors:

Martial Piotto (martial.piotto@bruker.fr)
François-Marie Moussallieh (fmmoussallieh@unistra.fr)
Agnes Neuville (neuville@bergonie.com)
Jean-Pierre Bellocq (jean.pierre.bellocq@chru-strasbourg.fr)
Karim Elbayed (elbayed@unistra.fr)
Izzie Jacques Namer (Izzie.Jacques.NAMER@chru-strasbourg.fr)

Version: 4 Date: 23 November 2011

Author's response to reviews: see over
Dear Editor,

Please find the revised version of our manuscript entitled:

**Towards real-time metabolic profiling of biopsy specimen during a surgical operation by $^1$H HRMAS-NMR: a case report**

We present the modifications requested by the editorial team.

1. We removed the hospital name and replaced by “our hospital”.
2. We included a discussion section as suggested.
3. We included the sentence “All authors read and approved the final manuscript” in the author’s contribution section.

All these modifications are highlighted in red in the text.

Yours sincerely,

I.J. Namer